Author's response to reviews

Title: Do men and women differ in physical functioning and response to mobility training after hip fracture: a secondary analysis of the Enhancing Mobility After Hip Fracture trial

Authors:

Lynda M Woodward (lynda.woodward@sydney.edu.au)
Lindy Clemson (lindy.clemson@sydney.edu.au)
Anne M Moseley (amoseley@georgeinstitute.org.au)
Stephen R Lord (s.lord@neura.edu.au)
Ian D Cameron (ian.cameron@sydney.edu.au)
Catherine Sherrington (csherrington@georgeinstitute.org.au)

Version: 2
Date: 16 August 2014

Author's response to reviews:

- addition of line numbers
- alteration in title: using sentence case instead of all capitals as per instructions